Author's response to reviews

Title: Randomised controlled Trial of Improvisational Music therapy's Effectiveness for children with Autism spectrum disorders (TIME-A): Study protocol

Authors:

Monika Geretsegger (monika.geretsegger@univie.ac.at)
Ulla Holck (holck@hum.aau.dk)
Christian Gold (christian.gold@grieg.uib.no)

Version: 2 Date: 17 November 2011

Author's response to reviews:

Dear Editors,

We are re-submitting our manuscript entitled "Randomised controlled Trial of Improvisational Music therapy's Effectiveness for children with Autism spectrum disorders (TIME-A): Study protocol" for consideration for publication in BMC Pediatrics.

The only change compared to the manuscript submitted on August 24, 2011, is the addition of a sentence in the Acknowledgements section on page 17, stating that the trial is funded by The Research Council of Norway, programme Mental Health (project number 213982).

Written approval of this grant including details of the amount of funding awarded to the study will be provided as soon as possible.

The amended manuscript has been seen and approved by all authors.

Thank you for considering our manuscript.

Sincerely,

Monika Geretsegger.